Allen, Pam, NMENV

From: Maestas, Ricardo, NMENV

Sent: Wednesday, November 18, 2015 2:06 PM

To: Allen, Pam, NMENV

Subject: FW: WIPP Information - For Call Today - Revised
Attachments: 1300 Meeting Action Items 050715.xlsx

From: Russ Patterson [mailto:russ.patterson@cbfo.doe.gov]

Sent: Tuesday, May 12, 2015 10:24 AM

To: Anthony Stone; Kliphuis, Trais, NMENV; 'peake.tom@epa.gov'; 'Edwards, Jonathan'; 'Walsh, Jonathan’; 'Perrin, Alan';
'Stone.Nick@epa.gov'; Smith, Coleman, NMENV; 'brozowski.george@epa.gov'; 'Fraass, Ron'; 'Hardy, Russell';
'Veal.Lee@epamail.epa.gov'; 'Economy.Kathleen@epa.gov'; 'Poppell.Sam@epa.gov'; Maestas, Ricardo, NMENV; 'Faller,
Scott H."; Holmes, Steve, NMENV; 'gene.turner@nnsa.doe.gov'; 'Karen.armijo@nnsa.doe.gov';
'laura.day@nnsa.doe.gov'; 'david.nickless@nnsa.doe.gov'; Peter Maggiore - LANL NNSA; 'Robinson, Bruce Alan
(robinson@lanl.gov)'; 'Sisk-Scott, Charlotte - WSMS (charlotte.sisk-scott@wipp.ws)'; LucasKamat, Susan, NMENV;
'Schreiber, Arleen Thorn (arleen@lanl.gov)'; 'Bignell, Dale - CTAC'

Cc: George Basabilvazo - WiPPNet; 'Reynolds, Tammy - NWP'; 'Harris, Alton - DOE EM'; 'Kennedy, Scott - NWP';
Anderson Ward; 'Jones, Stewart - RES'; 'Oates, Berta - CTAC'; 'schultheisz.daniel@epa.gov'; Philip Theisen; 'Kouba, Steve
- WRES'; Roger Nelson - WIPPNet; Susan McCauslin - WIPPNet; 'Pace, Berry'; J.R. Stroble - WIPPNet; 'Keffer, Sue’;
'christine.gibbs@nnsa.doe.gov'; 'Chavez, Rick - RES'; 'Dunagan, Sean - SNL'; Mark Pearcy - NWP; 'Salness, Rick - RES
(richard.salness@wipp.ws)'

Subject: RE: WIPP Information - For Call Today - Revised

May 12, 2015

The call in number is: 1-866-757-5307
Participant code: 5794706

Below is a summary of activities for today's call. (Please note that activities and dates are subject to change).

* Work activities (e.g., installation of substantial barrier, steel bulkhead and flashing) on the S-3080 drift for the
initial closure of Panel 6 is complete.

* Water spraying is continuing to be performed on the ribs, floor and ceiling of Rooms 1 through 6 of Panel 7.

* Work activities on the Room 7 closure of Panel 7 are progressing. The $-2520 drift room closure bulkhead is
installed and flashing is being installed.

* Excavation work activities for the interim ventilation fan/filter pads continue.

* The next Town Hall Meeting is scheduled, June 4th, 2015.

If you have any questions for this week's call please contact me or George Basabilvazo,

1 Wi



Russ Patterson



WIPP RECOVERY 1:00 WEEKLY MEETING ACTION LIST

Revised 5/07/15

A B C D E F
Date Requestor Action Responsible Person Due Date Status
1 | Identified |
2 7 WIPP-SPECIFIC ACTION ITEMS
Requested a copy of the detailed WIPP
Nick Stone, |Recovery Plan. Revised 4/21/15 -
11/04/14 NMED  |Requested a copy of the detailed Formerly item #120.
5 Primavera schedule. -
Requested a copy of the "Site
Implementation of the Recovery Plan"
11/25/14 | Coleman afte.r it receives DOE/HQ approval. Formerly Item # 121.
Smith Revised 3/3/15 - Requested the
| Corrective Action Plans that cover the
6 \Recovery Plan for Decontamination. |
. . ] Closed. Provided in the Nitrate Salt
02/10/15 Steve Requested a list of equipment that will be Suggs Bearing Waste Container Isolation Plan
7 Holmes leftin Panet7. Update distributd the week of 4/27/15.
Coleman . L ) ) ‘
8 03/10/15 Smith Provide a periodic status of Station A. Basabilvazo :Formerly Item # 147.
Requested the brand name(s) and model
Steve numbers of the three filter types used in
04/28/15 the main exhaust filter building, ’
Holmes . . . .
specifically, mod filter, high efficiency pre-
9 filter and HEPA filter.
05/05/15 Coleman [Requested results of MSHA monitoring in Jones
10 Smith the underground.




WIPP RECOVERY 1:00 WEEKLY MEETING ACTION LIST

Revised 4/22/15
A B C D E F
Date . ;
e Requestor Action Responsible Person | Due Date Status
Identified o S B
LANL ACTION ITEMS
Kathy If LANL cannot initiate a reactionin a LANL testing continues. Formerly Item
07/15/14 . LANL
Economy |MINO2 waste stream, what will be done? # 90.
3/31/15 - N. Stone requested this item
Requested that LANL representative be / / . a .
on next week’s call to provide a status remain open until the LANL testing
02/24/15 | Nick Stone P . LANL {tbd) results report is issued in the mid-
update on progress regarding drum .
> summer timeframe. Formerly Item #
breach determination.
— 145.
Revised 3/10/15 - How many SWBs
o /10/ L ¥ . K. Economy requested and B. Robinson
containing Swheat kitty litter and nitrate .
Coleman |salt (currently stored and undergoing agreed to provide a verbal summary
03/03/15 LANL, Robinson followed by a written document
103/ Smith  |HSG monitoring at LANL) have HSG | y

results that indicate a reaction is

ioccurring?

available some weeks later. Formerly
Item # 146.




